Author’s response to reviews

Title: Case Report: Histopathology of a Retrocorneal Membrane after Descemet Membrane Endothelial Keratoplasty

Authors:

Tarek Bayyoud (tarek_bayyoud@gmx.de)
Jens Rohrbach (martin.rohrbach@med.uni-tuebingen.de)
Karl Bartz-Schmidt (u.bartz-schmidt@med.uni-tuebingen.de)
Sebastian Thaler (sebastian.thaler@med.uni-tuebingen.de)

Version: 1 Date: 12 Dec 2018

Author’s response to reviews:

Point-by-point response with highlighted changes to the manuscript:

- Editor's comments:
  - The title of the case report was changed to be more descriptive as requested;
  
    - The histo-pathological examination’s result was extended as advised; an additional staining was performed.

We thank the Editor for helping us in significantly improving the article.

Reviewer 1

1. Quality of written English:

The manuscript was proof-read by a native speaker. We thank the reviewer for this point helping us to improve the manuscript.
Reviewer 2

1. Detailed description of the histological features of the membrane:

The histological description was better elucidated to highlight the originality of the article as recommended. In addition, a histological image was added.

2. Conclusion:

The conclusion section was adjusted as advised.

We thank the reviewer to improve the originality of the article.

Reviewer 3

We thank the reviewer for the positive feedback given.